Author's response to reviews

Title: Genomic profiling of rectal adenoma and carcinoma by array-based comparative genomic hybridization

Authors:

Zhi-Zhou Shi (zhizhoushi@gmail.com)
Yue-Ming Zhang (zhangyueming1028@126.com)
Jia-Jie Hao (hjj8173@126.com)
Tong-Tong Zhang (163zttong@163.com)
Bo-Shi Wang (wbs137@126.com)
Jian-Wei Liang (liangjianwei0@yahoo.com.cn)
Xi Chen (dejan11@sohu.com)
Ying Zhang (tilamisu-ji@163.com)
Gui-Qi Wang (wangguiq@126.com)
Ming-Rong Wang (wangmr07@gmail.com)
Yu Zhang (zhangyu909@126.com)

Version: 2 Date: 20 December 2011

Author's response to reviews:

Dear Editor-in-Chief,

We have made changes to this manuscript according to your advise. And we provide the point-by-point description of the changes as follows.

First, we added the section of “Patients and Samples” and introduced the information of samples used in this manuscript. All the samples used in this study were residual specimens after diagnosis sampling. And all patients signed separate informed consent forms for sampling and research. Second, we submitted our array CGH data to Gene Expression Omnibus with accession number GSE34472. We indicated this information at the end of section “Array-based CGH”. Third, we provided the clinical characteristics of the patients studied in this manuscript in Table 1.

Thank you for reading our work. We are looking forward to hearing from you.

Sincerely yours,

Ming-Rong WANG
State Key Laboratory of Molecular Oncology
Cancer Institute (Hospital), PUMC/CAMS
17 Panjiayuan Nanli, Chaoyang District
Beijing 100021, China.
Tel: 86-10-87788788; Fax: 86-10-87778651
E-mail: wangmr07@gmail.com